Author's response to reviews

Title: Assessment of lower urinary symptom flare with overactive bladder symptom score and International Prostate Symptom Score in patients treated with iodine-125 implant brachytherapy: Long-term follow-up experience at a single institute

Authors:

Makito Miyake (makitomiyake@yahoo.co.jp)
Nobumichi Tanaka (sendo@naramed-u.ac.jp)
Isao Asakawa (iasakawa@naramed-u.ac.jp)
Shunta Hori (horimaus@gmail.com)
Yosuke Morizawa (tigers.yosuke@gmail.com)
Yoshihiro Tatsumi (takuro.birds.nest@gmail.com)
Yasushi Nakai (nakaiyasusiuro@live.jp)
Takeshi Inoue (you1513tt@yahoo.co.jp)
Satoshi Anai (sanai@naramed-u.ac.jp)
Kazumasa Torimoto (torimoto@nmu-gw.naramed-u.ac.jp)
Katsuya Aoki (aokik@nmu-gw.naramed-u.ac.jp)
Masatoshi Hasegawa (hasegawa@naramed-u.ac.jp)
Tomomi Fujii (fujiiit@naramed-u.ac.jp)
Noboru Konishi (nkonishi@naramed-u.ac.jp)
Kiyohide Fujimoto (kiyokun@naramed-u.ac.jp)

Version: 4 Date: 26 Jul 2017

Author’s response to reviews:

We appreciate your considerable editorial work. We have decided to submit the revised manuscript to BMC Urology. Here is response to the comments.
1. Please provide a Keywords section after your Abstract that includes three to ten keywords representing the main content of the article.

> We added 6 key words after Abstract.

2. Thank you for providing a section on Ethics approval. We would appreciate it if you could please provide us with the full, official name of the committee which granted you approval for this study, and committee reference number if appropriate.

> ‘Nara Medical University Hospital Ethics Committee’ is the full, official name. The approval ID of this study is 1463. We have added the reference ID 1463 in the section on Ethics approval.

3. Thank you for submitting an Authors' contributions section. We had noticed you had included the initials "TY", yet these do not correspond to any authors listed for your manuscript. Please revise this section so that all authors correspond to one set of initials, or if an author is missing, please add them to the authors list.

> This is just a mistake. We have deleted ‘KT, KA and TY helped with writing manuscript’ from authors’ contribution section.

4. We had noticed you had listed 6 figures in your list of figure legends, but only 5 figures have been uploaded. Please ensure to include all figures with your manuscript.

> This is just a mistake. We have deleted the figure legend of Figure 6.